Author's response to reviews

Title: MicroRNA-125a is over-expressed in insulin target tissues in a spontaneous rat model of Type 2 Diabetes

Authors:

Blanca M Herrera (blanca@well.ox.ac.uk)
Helen E Lockstone (hel23@well.ox.ac.uk)
Jennifer M Taylor (jmtaylor@well.ox.ac.uk)
Quin F Wills (wills@stats.ox.ac.uk)
Pamela J. Kaisaki (pamela.kaisaki@well.ox.ac.uk)
Amy Barrett (amy.barrett@drl.ox.ac.uk)
Carme Camps (carme.camps@well.ox.ac.uk)
Cristina Fernandez (cristina.fernandez@well.ox.ac.uk)
Jiannis Ragoussis (oannisr@well.ox.ac.uk)
Dominique Gauguier (gdomi@well.ox.ac.uk)
Mark I McCarthy (mark.mccarthy@drl.ox.ac.uk)
Cecilia M Lindgren (celi@well.ox.ac.uk)

Version: 7 Date: 13 August 2009

Author's response to reviews:

Dear Editor,

The only change we would like to include to the manuscript is to add an affiliation for one of the senior authors. We would like to add the following affiliation to Dominique Gauguier:

INSERM U872, Centre de Recherche des Cordeliers, Rue de l'Ecole de Medecine, 75006 Paris, France.

And keep this as well as his Oxford affiliation.

Best wishes,

Blanca Herrera